Author’s response to reviews

Title: Mobile phone applications to overcome malnutrition among preschoolers: a systematic review

Authors:

Navisa Seyyedi (navisa_seiedi67@yahoo.com)
Bahlol Rahimi (bahlol.rahimi@gmail.com)
Hamid Reza Farrokh Esamlou (hamidfarrokh@gmail.com)
Toomas Timpka (toomas.timpka@liu.se)
Hadi Lotfnejhad Afshar (hadi.afshar@gmail.com)

Version: 1 Date: 26 Jul 2018

Author’s response to reviews:

Dear Editor,

We highly appreciate your detailed comments on our manuscript “MIDM-D-18-00217R1”. The suggestions were most helpful for us and we have incorporated them in the revised paper.

As below, on behalf of my co-authors, I would like to clarify some of the points raised by you. We hope that you will be satisfied with our responses to the comments and the revisions on the original manuscript.

Thanks and Best Regards!

Bahlol Rahimi

2018-07-27

BMC Medical Informatics and Decision Making

Manuscript No. MIDM-D-18-00217R1

Editor's comments:
We note that the current submission contains some textual overlap with other previously published works, in particular:


Response: The textual overlap in the sentences have been rephrased.


Response: The textual overlap in the sentences have been rephrased.


Response: The textual overlap in the sentences have been rephrased.


Response: The textual overlap in the sentences have been rephrased.

This overlap mainly exists in the Results. While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Response: In the revised version, the said sentences have been rephrased in order to remove textual overlap in the sentences.